Author's response to reviews

Title: Secondary breast lymphoma diagnosed by Vacuum-assisted breast biopsy: case report.

Authors:

Flora Zagouri (florazagouri@yahoo.co.uk)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Afrodit Nonni (afnonni@med.uoa.gr)
Dimitra Koulocheri (dim.koul@altecornet.gr)
Philip Domeyer (philip.domeyer@gmail.com)
Dimitrios Dardamanis (dardaman@hol.gr)
Nikolaos V. Michalopoulos (nmichal@med.uoa.gr)
Nikolaos Pararas (npararas@med.uoa.gr)
Antonia Gounaris (antgouna@otenet.gr)
George C. Zografo (gzografo@med.uoa.gr)

Version: 2 Date: 14 June 2007

Author's response to reviews: see over
Dear Editor,

I am sending for publication to your distinguished Journal the revised version of the case report entitled: “Secondary breast lymphoma diagnosed by Vacuum-assisted breast biopsy: case report.”

According to the Assistant Editor’s comments:
1. In the Competing interests section, the sentence "The authors declare that they have no competing interests." has been added.
2. Within the Acknowledgements section, it is stated that written informed patient consent was obtained for publication
3. The reference list has been cut down to 15 references.

Yours sincerely,

George C. Zografos,
Assoc. Professor of Surgery
University of Athens